Title: Ruptured Aneurysm at the Fenestration of the Middle Cerebral Artery Detected by Magnetic Resonance Angiography in a Patient with Systemic Lupus Erythematosus and Renal Failure: a case report

Authors:

Sadaharu Tabuchi (tabuchis@pref.tottori.jp)
Hiroki Yoshioka (h-yoshioka@hotmail.co.jp)

Version: 3 Date: 21 November 2013

Author's response to reviews: see over
The Biomed Central Editorial Team

Object: MS: 3637610811119170
Ruptured Aneurysm at the Fenestration of the Middle Cerebral Artery Detected by Magnetic Resonance angiography in a Patient with Systemic Lupus Erythematosus and Renal Failure. Dr Sadaharu Tabuchi et al.

Thank you for consideration of our manuscript for publication in your journal. The manuscript was immediately corrected along with your suggestions in November 21. We would like to express our sincere thanks for these important comments. Points of the revision and responses to your suggestions are summarized as follows;

1. Please include the Ethnicity of the patient in the Abstract and Case presentation sections.

I added “Japanese” in patient information in the Abstract and Case presentation sections.

2. Please include a list of abbreviations used in the manuscript and their meanings after the Conclusions section.

I added a list of abbreviations in page 4 as follows;

**Abbreviations**

MCA: middle cerebral artery; SAH: subarachnoid hemorrhage; SLE: systemic lupus erythematosus; MRA: Magnetic resonance angiography; VP: ventriculo-peritoneal; 3D-TOF: three-dimensional time-of-flight; VR: volume rendering; ICG: indocyanine green; GFR: glomerular filtration rate; CKD: chronic kidney disease; MIP: maximum intensity projection; DSA: digital subtraction angiography
3. Please include an acknowledgement section at the end of the manuscript before the reference list. Please acknowledge anyone who contributed towards the study by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content, but who does not meet the criteria for authorship. Please also include the source(s) of funding for all authors. Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements.

I added an acknowledgement section in page 4 as follows;

Acknowledgement

We thank Dr. Eiji Matsusue (Department of Radiology) for providing us an excellent imaging data, Dr. Toshihiko Masago (Department of Urology) and Dr. Hiromi Omura (Department of Internal medicine, Tottori Prefectural Central Hospital) for supporting the management of general conditions.

New corrections were presented in violet color letter in the final revised manuscript.

We hope that the revised manuscript is now suitable for publication of “Journal of Medical Case Reports”.
I do look forward to hearing from you a constructive response at your earliest convenience.

Sincerely yours,

Sadaharu Tabuchi M.D., Ph.D.